RESEARCH



Sex differences in obesity-induced renal lipid accumulation revealed by lipidomics: a role of adiponectin/AMPK axis

Florian Juszczak^{1,2*}, Louise Pierre^{1,3}, Morgane Decarnoncle¹, Inès Jadot², Blanche Martin², Olivia Botton², Nathalie Caron², Jonas Dehairs⁴, Johannes V. Swinnen⁴ and Anne-Emilie Declèves¹

Abstract

Background Sex differences have been observed in the development of obesity-related complications in patients, as well as in animal models. Accumulating evidence suggests that sex-dependent regulation of lipid metabolism contributes to sex-specific physiopathology. Lipid accumulation in the renal tissue has been shown to play a major role in the pathogenesis of obesity-induced kidney injury. Unlike in males, the physiopathology of the disease has been poorly described in females, particularly regarding the lipid metabolism adaptation.

Methods Here, we compared the lipid profile changes in the kidneys of female and male mice fed a high-fat diet (HFD) or low-fat diet (LFD) by lipidomics and correlated them with pathophysiological changes.

Results We showed that HFD-fed female mice were protected from insulin resistance and hepatic steatosis compared to males, despite similar body weight gains. Females were particularly protected from renal dysfunction, oxidative stress, and tubular lipid accumulation. Both HFD-fed male and female mice presented dyslipidemia, but lipidomic analysis highlighted differential renal lipid profiles. While both sexes presented similar neutral lipid accumulation with obesity, only males showed increased levels of ceramides and phospholipids. Remarkably, protection against renal lipotoxicity in females was associated with enhanced renal adiponectin and AMP-activated protein kinase (AMPK) signaling. Circulating adiponectin and its renal receptor levels were significantly lower in obese males, but were maintained in females. This observation correlated with the maintained basal AMPK activity in obese female mice compared to males.

Conclusions Collectively, our findings suggest that female mice are protected from obesity-induced renal dysfunction and lipotoxicity associated with enhanced adiponectin and AMPK signaling compared to males.

Highlights

- HFD female mice are protected from ectopic lipid accumulation in the renal tubules compared to males, despite similar body weight gain.
- · Lipidomic analysis revealed sex differences in renal lipid profiles.

*Correspondence: Florian Juszczak Florian.juszczak@umons.ac.be Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/.

- HFD female mice are protected from renal phospholipid accumulation and lysosomal dysfunction compared to HFD male mice.
- HFD males specifically present increased levels of very-long-chain ceramides in the kidney and decreased AMPK activity.
- HFD females may be protected from renal injury by higher levels of adiponectin, which correlates with higher AMPK activity in the kidney.

Keywords Sex difference, Chronic kidney disease, Obesity, Renal lipids, Adiponectin, AMPK

Plain language summary

Obesity-related complications can differ between men and women due to sex-specific differences in how fats are handled. Here, we studied the effects of high-fat diet on the kidneys of male and female mice. We found that despite gaining similar weight, obese female mice were better protected against insulin resistance, liver fat accumulation, and kidney damage caused by obesity than males. In particular, female mice were protected against lipid accumulation in the kidneys. We further analyzed the lipid profile in the kidneys of both male and female mice and observed differences in the amount and nature of the accumulated lipids. Male mice had increased levels of specific lipids, which may contribute to their higher risk of kidney damage. In contrast, female mice showed better lipid metabolism adaptation, which helped to protect their kidneys. This study also revealed an association between higher levels of the protein hormone adiponectin and higher activity of the cellular energy master regulator protein AMPK in obese females. These proteins may help prevent obesity-induced kidney damage. In obese males, these protective proteins are reduced and are associated with kidney damage and lipid accumulation in the kidneys. Obesity in males is associated with the presence of potentially toxic lipids and dysregulated renal metabolism. Understanding these sex-related differences in obesity-related complications could lead to better management and treatment of kidney problems in both men and women.

Introduction

The incidence of obesity has dramatically increased over the last few years and is expected to reach 50% by 2030 [1]. Obesity is a major risk factor for chronic kidney disease (CKD), and is the second most important predictor of end-stage renal disease [2]. Ectopic lipid accumulation in proximal tubular cells (PTC) have also been observed, indicating that lipotoxicity play a role in the development and progression of CKD [3]. Notably, sex differences have been observed in the development of obesity and obesity-related complications in both animal models and patients [4-8]. Men and women differ in the underlying pathophysiology of CKD and its associated complications [9]. Epidemiological studies on the global burden of CKD highlighted a higher prevalence in women, whereas mortality is higher in men [10]. Although recent studies have provided some insights into the sexual dimorphism of obesity-induced CKD, there remains a critical need to gain a better understanding of the physiological sex-specific characteristics of this disease.

Mice fed a HFD have been widely used to study obesity-related disorders [11]. We and others have previously demonstrated the renal consequences of obesity in this animal model [2, 12–16]. HFD-fed male mice developed human characteristic features of obesity-induced CKD including glomerulosclerosis, tubulointerstitial fibrosis associated with albuminuria and proteinuria [17]. In addition, obesity-induced CKD is characterized by ectopic lipid depositions in renal PTC both in rodents and humans, leading to renal lipotoxicity [12-14, 18-20]. Indeed, dysregulated renal lipid metabolism, particularly fatty acid oxidation impairment, has been discovered to play a major role in CKD development and progression [21]. Particularly, accumulation of lysosomal phospholipids was demonstrated in PTC of obese male mice, which is associated with lysosomal dysfunction, impaired autophagic flux, mitochondrial dysfunction, and inflammasome activation [13, 22, 23]. Even if the renal consequences of obesity are well characterized in males, to date, there are still no reliable data in females. To note, we and others demonstrated the pivotal role of AMPK in the development of obesity-induced CKD [24-29]. AMPK is a ubiquitous heterotrimeric kinase considered as an energetic sensor responding to changes in the intracellular AMP/ATP ratio [30]. Metabolic stresses impairs AMPK activity in the renal cortex of obese and diabetic male mice [3, 25]. In addition, AMPK activation prevents renal inflammation, oxidative stress and improves

mitochondrial dysfunction in several male animal models of CKD [3, 14, 31, 32]. Moreover, adiponectin has been shown to enhance AMPK activity in various tissues, including the kidney [33, 34]. However, in diabetes, adiponectin production is decreased, leading to impaired AMPK activity in the renal cortex of diabetic male mice [26, 35]. Adiponectin deficiency has been associated with the development of insulin resistance and metabolic dysfunction in various tissues, including the kidney. Adiponectin activates AMPK and promotes fatty acid oxidation [36]. Studies have shown that females tend to have higher levels of adiponectin compared to males, which may contribute to sex differences in lipid metabolism [37, 38]. Given the limited understanding of the effects of obesity on CKD in females, the present study seeks to comprehensively characterize the sex-specific consequences of obesity on kidney function in HFD-fed mice. Here, we hypothesized that HFD triggers differential accumulation of lipid species in the kidneys of male and female mice, explaining the differences in the development and progression of obesity-induced CKD. In this study, we used targeted lipidomic analysis to quantify the sexual dimorphism in renal lipid accumulation. Moreover, the AMPK-adiponectin axis was investigated to further highlight its implications in renal lipid metabolism and CKD outcomes.

Materials and methods

Animals

The study conformed to the APS Guidelines for the Care and Use of Animals and was approved by the Animal Ethics Committee of the University of Namur. The experiments were conducted on C57Bl/6J male and female mice (Janvier Labs, Le Genest Saint-Isle, France) that were housed in cages with free access to food and water. The mice were maintained at 35-40% relative humidity and a temperature of 20-23 °C with a 12:12 h light-dark cycle. Over a 16-week period, 8-week-old mice (n=6)per sex and per treatment) were randomized to either a LFD (10% of total calories from fat; D12450J, Research Diets, New Brunswick, NJ, USA) or HFD (60% of total calories from fat; D12492, Research Diets, New Brunswick, NJ, USA). Body weight (BW) was measured every 4 weeks and 24-h urine collection was performed at the end of the study using metabolic cages. Fasting blood glucose concentration (One Touche Vita, LifeScan Inc., Milpitas, CA) was measured from the tail vein. At week 16, mice were anesthetized after an overnight fast with a solution of ketamine (Nimatek[®], Eurovet Animal Health, Blabel, The Netherlands, 80 mg/kg b.w.) and medetomidine (Domitor[®], Orion Pharma, Espoo, Finland, 0.5 mg/ kg b.w.), and blood was collected by intracardiac puncture. Blood samples were collected and centrifuged at a high speed for 20 min at 4 °C. The plasma was collected and stored at -80 °C until further use. The kidneys, liver, and heart were removed, weighed, and reported to tibia length in order to normalize the data, as previously reported [14]. A portion of the collected organs was fixed in Duboscq-Brazil solution, while the remaining tissues were frozen in liquid nitrogen and stored at -80 °C for further analysis.

Urine collection and urinary markers analyses

After 16 weeks of feeding, mice were placed for 24-h urine collection in metabolic cages that provide ad libitum access to food and water and allow refrigeration of urine samples. Urine was subsequently stored at -20 °C. Urinary albumin and creatinine levels were measured using a mouse Albuwell ELISA kit and Creatinine Companion kit (Exocell, Philadelphia, PA, USA). Total proteinuria was quantified using the Bradford method based on the absorbance of the Coomassie Brilliant Blue dye. As an index of oxidative stress, urine samples were analyzed for hydrogen peroxide using the Amplex red assay (Thermo Fisher Scientific, Waltham, MA, USA) following the manufacturer's instructions. All urinary marker values were normalized to the urinary creatinine concentration.

Biochemical assays

Plasma leptin and adiponectin (full-length form of adiponectin) concentrations were determined by ELISA (Mouse/rat leptin immunoassay Quantikine ELISA, Mouse adiponectin/Acrp30 Immunoassay Quantikine ELISA, R&D Systems Europe, Abingdon, UK). Plasma insulin levels were determined by ELISA using the rat/ mouse insulin ELISA kit (Merck, Darmstadt, Germany). The homeostasis model assessment (HOMA-IR) for the insulin resistance index was determined using a calculator available from the Oxford Center for Diabetes, Endocrinology, and Metabolism (https://www.dtu.ox.ac.uk/ homacalculator/). Colorimetric enzymatic tests were performed to measure plasma cholesterol levels (Diasys, Diagnostic System, Holzheim, Germany) and plasma non-esterified fatty acid (NEFA) levels (Wako Pure Chemical Industries, Ltd., Osaka, Japan), following the manufacturer's instructions.

Histology and morphological analyses

Five-µm paraffin-embedded kidney sections were stained with Periodic Acid Schiff (PAS), Hemalun, and Luxol Fast Blue to assess morphological alterations. Morphometry of kidney sections was performed as previously reported [13]. Briefly, the frequency of tubules containing vacuolated cells was evaluated using a semi-quantitative singleblind analysis. To standardize the evaluation procedure, an additional lens engraved with a square grid was inserted into one of the microscope's eyepieces. For each paraffin section, 10 square fields (0.084 mm²/field) were observed at \times 400 magnification. Ten randomly selected areas of each cortex kidney section were analyzed using the ImageJ software. Paraffin-embedded liver sections were stained with hematoxylin and eosin and steatosis was graded as described by Ryu et al. [39].

Immunohistochemistry

Five-µm paraffin-embedded kidney sections were dewaxed and rehydrated, followed by microwave pre-treatment in 1 mM EDTA buffer to unmask antigens present in the renal tissue. Endogenous peroxidase activity was removed by incubation with 3% H₂O₂ for 10 min and blocking with 10% normal goat serum. Sections were incubated with a primary antibody against LAMP-1 (Abcam, Cambridge, UK) overnight at 4 °C. After rinsing in TBS, slides were exposed for 30 min with SignalStain® Boost IHC Detection Reagent (Cell Signaling, Danvers, MA, USA), and bound peroxidase activity was detected using a DAB kit (Agilent DAKO, Heverlee, Belgium). Counterstaining was performed using Hemalun and Luxol Fast Blue. The evaluation of the relative positive area was performed on one section per experimental animal. For each section, ten square fields $(0.084 \text{ mm}^2/$ field) were observed at 400×magnification in each renal zone. The relative area occupied by positive staining was expressed as a percentage.

Quantitative real-time polymerase chain reaction (PCR)

Frozen kidney cortex was homogenized and total RNA was extracted using TRIzol (Sigma-Aldrich, St. Louis, MO, USA) and treated with DNAse (Promega, Madison, WI, USA). The total RNA concentration was measured using a NanoDrop spectrophotometer (NanoDrop 1000, Thermo Fisher Scientific, Waltham, MA, USA). Transcript-specific primers were generated based on the mouse sequences from GenBank. The NCBI Primer BLAST was used to ensure the specificity of the primers for each target. All primer pairs were analyzed for their dissociation curves and melting temperatures. Real-time quantitative PCR was performed to quantify the mRNA levels of AdiporR1, Lamp1, Cathepsin D, p62, CerS2, CerS5, CerS6, Acer2, Acer3, and 18S as housekeeping gene (Table 1). Briefly, 2 µg of total RNA was reversetranscribed using MLV reverse transcriptase (Promega, Madison, WI, USA) for 1 h at 70 °C. Quantitative PCR amplification was performed using SYBR Green Master Mix (Roche, Belgium) and Prism 7300 Real-Time PCR Detection System (Applied Biosystems, CA, USA). Mean fold changes were calculated by averaging duplicate measurements for each gene. The relative gene expression was calculated using the $2^{-\Delta\Delta CT}$ method.

 Table 1
 Primer
 sequences
 for
 RT-qPCR
 analysis
 of
 mRNA

 expression

 </td

Gene		Primer sequences (5'-3')
CerS2	Fw	ATGCTCCAGACCTTGTATGACT
	Rv	CTGAGGCTTTGGCATAGACAC
CerS5	Fw	CGGGGAAAGGTGTCTAAGGAT
	Rv	GTTCATGCAGTTGGCACCATT
CerS6	Fw	GATTCATAGCCAAACCATGTGCC
	Rv	AATGCTCCGAACATCCCAGTC
Acer2	Fw	GTGTGGCATATTCTCATCTG
	Rv	TAAGGGACACCAATAAAAGC
Acer3	Fw	TGTGATTCACTGAGGAACTTTCG
	Rv	AGAAACTTCACTTTTGGCCTGTA
AdipoR1	Fw	TTTGCCACTCCCAAGCAC
	Rv	ACACCACTCAAGCCAAGTCC
Lamp1	Fw	CAGCACTCTTTGAGGTGAAAAAC
	Rv	ACGATCTGAGAACCATTCGCA
CathepsinD	Fw	CTCAAAGGCCCCATCACCAA
	Rv	TGCCGTTCTTCACATAGG
p62	Fw	GATGTGGAACATGGAGGGAAGAG
	Rv	AGTCATCGTCTCCTCCTGAGCA

Western blot analysis

Proteins were extracted from renal cortex tissues using Cell Lysis Buffer (Cell Signaling, Danvers, MA, USA) with phosphatase and protease inhibitor cocktail (Thermo Fisher Scientific, Waltham, MA, USA) at 4 °C followed by centrifugation at 14,000×g for 15 min at 4 °C. Protein concentrations were quantified by Pierce BCA assay kit (Thermo Fisher Scientific, Waltham, MA, USA) and then 20 µg of total lysate were separated by SDS-PAGE 12% and transferred onto nitrocellulose membranes. Following blocking step in 5% BSA for 1 h, the membranes were incubated with primary antibodies against phosphorylated AMPK (P-AMPK), AMPK (Cell Signaling, Danvers, MA, USA), AdipoR1 (Abcam, Cambridge, UK) or β-actin (Thermo Fisher Scientific, Waltham, MA, USA) overnight at 4 °C and then with secondary antibodies (Li-Cor Biosciences, Lincoln, NE, USA) for 1 h at room temperature. Antibodies were diluted in Odyssey Blocking Buffer TBS containing 0.1% Tween20. Proteins were visualized and quantified using the Odyssey® imaging system (Li-Cor Biosciences, Lincoln, NE, USA).

Lipidomic analysis

Lipid extraction Tissue lysates containing 10 μ g of DNA were homogenized in 700 μ l of water with an handheld sonicator and were mixed with 800 μ l HCl(1 M):CH3OH 1:8 (v/v), 900 μ l CHCl3, 200 μ g/ml of the antioxidant 2,6-di-tert-butyl-4-methylphenol

(BHT; Sigma-Aldrich, St. Louis, MO, USA) and 3 μ l of SPLASH[®] LIPIDOMIX[®] Mass Spec Standard (#330707, Avanti Polar Lipids, Birmingham, AL, USA). After vortex and centrifugation, the lower organic fraction was collected and evaporated using a Savant Speed-vac spd111v (Thermo Fisher Scientific, Waltham, MA, USA) at room temperature and the remaining lipid pellet was stored at – 20 °C under argon.

Mass spectrometry (MS) Lipid pellets were reconstituted in 100% ethanol and analyzed by liquid chromatography electrospray ionization tandem mass spectrometry (LC-ESI/MS/MS) on a Nexera X2 UHPLC system (Shimadzu) coupled with hybrid triple quadrupole/linear ion trap mass spectrometer (6500 + QTRAP system; AB SCIEX). Chromatographic separation was performed on a XBridge amide column (150 mm \times 4.6 mm, 3.5 μ m; Waters) maintained at 35 °C using mobile phase A [1 mM ammonium acetate in water-acetonitrile 5:95 (v/v)] and mobile phase B [1 mM ammonium acetate in water-acetonitrile 50:50 (v/v)] in the following gradient: (0–6 min: 0% B \rightarrow 6% B; 6–10 min: 6% B \rightarrow 25% B; 10–11 min: 25% B \rightarrow 98% B; 11–13 min: 98% B \rightarrow 100% B; 13–19 min: 100% B; 19-24 min: 0% B) at a flow rate of 0.7 ml/min which was increased to 1.5 ml/min from 13 min onwards. Sphingomyelins (SM), cholesterol esters (CE) and ceramides (CER) were measured in positive ion mode with a precursor scan of 184.1, 369.4, 264.4, respectively. Triacylglycerides (TG), diacylglycerides (DG) and monoacylglycerides (MG) were measured in positive ion mode with a neutral loss scan for one of the fatty acyl moieties. Phosphatidylcholines (PC), phosphatidylethanolamines (PE), phosphatidylglycerols (PG), phosphatidylinositols (PI) and phosphatidylserines (PS) were measured in negative ion mode by fatty acyl fragment ions. Lipid quantification was performed by scheduled multiple reactions monitoring, the transitions being based on the neutral losses or the typical product ions as described above. The instrument parameters were as follows: Curtain Gas = 35 psi; Collision Gas = 8 a.u. (medium); IonSpray Voltage = 5500 V and - 4500 V; Temperature = 550 °C; Ion Source Gas 1 = 50 psi; Ion Source Gas 2=60 psi; Declustering Potential=60 V and -80 V; Entrance Potential = 10 V and -10 V; Collision Cell Exit Potential = 15 V and - 15 V.

The following fatty acyl moieties were taken into account for the lipidomic analysis: 14:0, 14:1, 16:0, 16:1, 16:2, 18:0, 18:1, 18:2, 18:3, 20:0, 20:1, 20:2, 20:3, 20:4, 20:5, 22:0, 22:1, 22:2, 22:4, 22:5 and 22:6 except for TGs which considered: 16:0, 16:1, 18:0, 18:1, 18:2, 18:3, 20:3, 20:4, 20:5, 22:2, 22:3, 22:4, 22:5, 22:6.

Data analysis Peak integration was performed with the MultiQuantTM software version 3.0.3. Lipid species

signals were corrected for isotopic contributions (calculated with Python Molmass 2019.1.1), quantified based on internal standard signals and adheres to the guidelines of the Lipidomics Standards Initiative (LSI) (level 2 type quantification as defined by the LSI).

Statistical analysis

The results are presented as the mean \pm SEM. The level of statistical significance was set at p < 0.05. Analyses were performed using Prism GraphPad Software version 6 (San Diego, CA, USA). Differences were analyzed using two-way ANOVA to examine the overall effects of diet, sex, and interaction (see detailed statistical analysis in Table 2). In cases of significant ANOVA effects, post hoc comparisons were performed using Tukey's multiple comparisons test to determine the significance between groups when appropriate. For lipidomic data, multiple t-tests with *p*-values corrected for multiple comparisons using the Bonferroni method were performed within each sex-matched group to analyze the FA profile for each lipid class. Partial least squares discriminant analysis (PLS-DA) and heat map visualization were performed using MetaboAnalyst 4.0 (http://www.metaboanalyst.ca).

Results

Sexual dimorphism in metabolic parameters in response to HFD

To determine the differential effects of sex on dietinduced obesity, 8-week-old male and female C57Bl/6J mice were fed either a HFD or a LFD for 16 weeks. BW was measured over time and is shown in Fig. 1A as the relative BW gain (in % of the corresponding BW at the beginning of the protocol) to overcome sex differences. A similar relative increase in BW was measured in both the female and male HFD feeding groups throughout the experimental protocol. Both sexes also presented identical BW gains in the LFD-fed groups. Moreover, plasma leptin concentration was evaluated as a marker of dietinduced obesity. Leptin is an adipokine that correlates with body fat mass and increases with obesity [40]. Fig. 1B reveals a strong impact of the HFD on leptin concentration, as demonstrated by the two-way ANOVA analysis $(P_{\text{Diet}} < 0.0005)$. Moreover, post hoc comparisons revealed that plasma leptin levels were significantly higher in both HFD-fed male and female mice than in LFD-fed mice. However, the plasma leptin levels were significantly lower in females than in males (P_{Sex} < 0.0005). HFDinduced obesity in mouse models has been associated with insulin resistance. Therefore, we investigated the relationship between sex, diet, and glucose metabolism in HFD-fed mice. As illustrated in Fig. 1C, D, HFD-fed females were not affected by hyperglycemia and hyperinsulinemia compared with males. HOMA-IR calculation

Figure#	Measurement	Diet			Sex			Interaction			
		DFn, DFd	ш	d	DFn, DFd	н	d	DFn, DFd	н	d	
	Plasma leptin (ng/mL)	1.20	64.52	< 0.0001	1.20	8.31	0.0092	1.20	3.98	0.0597	
-	Fasting blood glucose (mg/dL)	1.20	14.15	0.0004	1.20	18.02	0.0012	1.20	2.35	0.1409	
-	Plasma insulin level (ng/mL)	1.17	8.17	0.0109	1.17	1,74	0.2043	1.17	1.17	0.2938	
-	HOMA-IR	1.16	27.11	< 0.0001	1.16	13.79	0.0019	1.16	6.75	0.0194	
-	Plasma FFA (mg/dL)	1.17	26.63	< 0.0001	1.17	4.36	0.0521	1.17	0.53	0.4762	
1	Plasma cholesterol (mg/dL)	1.18	25.22	< 0.0001	1.18	3.41	0.0813	1.18	0.26	0.6131	
-	Hepatic steatosis score	1.20	43.44	< 0.0001	1.20	7.98	0.0105	1.20	5.99	0.0237	
2	UACR (µg/mg Cre)	1.17	12.82	0.0023	1.17	0.39	0.5406	1.17	0.24	0.6305	
2	Urinary protein (mg/ mg Cre)	1.17	9.43	0.0069	1.17	18.53	0.0005	1.17	2.97	0.1028	
2	Urinary H ₂ O ₂ (nmol/ mg Cre)	1.20	19.73	0.0003	1.20	49.84	< 0.0001	1.20	10.82	0.0037	
5	Vacuolated tubules (/ mm ²)	1.20	16.36	0.0006	1.20	11.55	0.0029	1.20	10.99	0.0035	
9	% of LAMP1 ⁺ area	1.20	10.98	0.0044	1.20	12.61	0.0027	1.20	6.94	0.0181	
9	LAMP1 mRNA expres- sion	1.20	8.19	0.0093	1.20	23.69	< 0.0001	1.20	1.95	0.0177	
9	CathepsinD mRNA expression	1.20	8.36	0.0087	1.20	12.48	0.0020	1.20	0.28	0.6008	
9	p62 mRNA expression	1.20	5.74	0.0260	1.20	37.47	< 0.0001	1.20	0.19	0.6700	
7	CerS2 mRNA expression	1.18	7.07	0.0160	1.18	44.36	< 0.0001	1.18	2.626	0.1225	
7	CerS5 mRNA expression	1.19	23.85	0.0001	1.19	29.48	< 0.0001	1.19	0.01	0.9307	
7	CerS6 mRNA expression	1.19	25.34	< 0.0001	1.19	46.65	< 0.0001	1.19	1.702	0.2076	
7	Acer2 mRNA expression	1.19	18.97	0.0004	1.19	8.23	0.0098	1.19	3.93	0.0622	
7	Acer3 mRNA expression	1.19	12.83	0.002	1.19	17.89	0.0005	1.19	1.09	0.3086	
8	P-AMPK to AMPK ratio	1.19	9.29	0.0066	1.19	66.32	< 0.0001	1.19	3.51	0.0764	
00	Plasma adiponectin (µg/mL)	1.20	0.16	0.6946	1.20	12.67	0.0020	1.20	0.84	0.3717	
8	AdipoR1/β-actin	1.19	1.145	0.2981	1.19	5.521	< 0.0001	1.19	0.2969	0.0822	
œ	AdipoR1 mRNA expres- sion	1.16	6.12	0.0242	1.16	21.36	0.0002	1.16	4.5	0.0490	

 Table 2
 Results from two-way ANOVA statistical analysis

The table provides degrees of freedom, F and p scores for two-way ANOVA analyses. Significant p-values are displayed in blod



Fig. 1 Sex differences in metabolic parameters in response to HFD. Male and female C57BL/6J mice were fed with either HFD or LFD for 16 weeks. **A** Relative increase in BW. Statistical analyses were performed using two-way ANOVA followed by Tukey post-test. *p < 0.05 HFD vs LFD male and $p^{\pm} < 0.05$ HFD vs LFD female. n = 6 in each group. **B** Plasma leptin level. **C** Fasting blood glucose and **D** plasma insulin levels were assessed in male and female mice fed a LFD or a HFD at week 16. **E** HOMA-IR index calculation. **F** Plasma FFA and **G** cholesterol levels. **H** Representative photomicrographs (original magnification ×400) of H&E staining illustrating hepatic steatosis from liver sections in males and females fed a LFD or a HFD. **I** Semiquantitative grading of hepatic steatosis in each group. Statistical analyses were performed using two-way ANOVA followed by Tukey post-test. *p < 0.05; **p < 0.01; ***p < 0.001. Data are presented as means ± SEM

revealed that HFD induced insulin resistance in males (5.52 ± 0.67) , but not in females (2.87 ± 0.35) with a significant sex \times diet interaction (P_{Int} < 0.05) (Fig. 1E). Moreover, plasma free FA (FFA) and cholesterol concentrations were similarly increased in both the male and female HFD-fed groups (Fig. 1F, G). The results of the two-way ANOVA indicated a main effect of diet ($P_{Diet} < 0.05$) on plasma lipid concentrations, but no sex difference was observed. Because hepatic steatosis is a reliable marker of metabolic disorders in obesity models, we evaluated the liver steatosis score in all experimental groups (Fig. 1H, I). Significant lipid accumulation in the liver tissue was observed in HFD-fed male and female mice, as illustrated in Fig. 1H. Interestingly, HFD-fed female mice presented only a moderate increase in the hepatic steatosis score compared to males (Fig. 1I) with a significant $sex \times diet$ interaction (P_{Int} < 0.05), suggesting differential lipid accumulation in HFD-fed mice related to sex.

Sex difference in renal function and structure in response to HFD

Obesity-induced CKD is characterized by ectopic lipid accumulation in the S1 segment of PTC, leading to renal dysfunction [13]. Therefore, we first evaluated renal function by measuring urinary albumin (albuminuria) and protein (proteinuria) levels, both normalized to creatinine. In addition, urinary H₂O₂ level was measured as a marker of oxidative stress. As presented in Fig. 2A-C, albuminuria, proteinuria, and urinary H₂O₂ levels significantly increased in HFD-fed male mice. In female mice, only a slight increase (p=0.072) in albuminuria was observed (Fig. 2A). Two-way ANOVA revealed that albuminuria was mostly affected by diet rather than sex (Table 2), whereas proteinuria and urinary H_2O_2 levels were significantly affected by sex with a significant sex × diet interaction for urinary H_2O_2 ($P_{Int} < 0.005$). Morphological changes in the renal cortex were also examined. The effects of sex and diet on proximal tubule morphology were analyzed (Fig. 2D). As expected, HFDfed male mice displayed a high amount of vacuolized proximal tubules, as indicated by intracellular lipid droplet accumulation in proximal tubular cells. Surprisingly, female mice did not present any intracellular lipid droplets in the PTC when fed a HFD. These observations were confirmed by the quantification of the number of vacuolized tubules in each group (Fig. 2E). Two-way ANOVA analysis identified the effects of sex and diet, as well as a significant sex×diet interaction (P_{Int} < 0.005), suggesting a sex-specific effect on this parameter in HFD-fed mice.



Fig. 2 Histological changes and renal function in male and female mice fed a HFD. **A** Quantitative measurement of urinary albumin to creatinine ratio (UACR), **B** proteinuria and **C** urinary H_2O_2 level to creatinine ratio at week 16. **D** Representative photomicrographs (original magnification × 400) of PAS staining illustrating vacuolated proximal convoluted tubular cells from renal cortex section in males and females fed a LFD or a HFD. **E** Quantitative analysis of number of vacuolated tubules per mm². Statistical analyses were performed using two-way ANOVA followed by Tukey post-test. *p < 0.05; **p < 0.01; **p < 0.001. Data are presented as means ± SEM

Sex-specific lipid accumulation in response to a HFD in the renal cortex

To further characterize the sex differences observed in renal lipid accumulation, targeted lipidomic analysis using MS was performed on renal tissues from males and females fed a LFD or HFD. As illustrated in Fig. 3A, the heat map presenting quantitative levels of lipid species (SM, CE, CER, TG, DG, MG, and phospholipids) for





each animal demonstrates a strong accumulation of lipids in males fed a HFD for all lipid classes, whereas females seemed to be less affected by renal lipid accumulation following HFD feeding, which further confirms the absence of tubular vacuolization in HFD-fed females. To obtain an overview of the lipid species composition differences between sexes and dietary interventions, a PLS-DA was performed based on the relative amounts of lipid species for each animal. LFD and HFD-fed male mice formed separate clusters of lipid species, indicating different renal lipid compositions in response to diet, whereas HFD and LFD-fed female mice were not discriminated (Fig. 3B). Moreover, LFD-fed males and females presented separated clusters, suggesting a sex-specific composition of renal lipid species.

Effect of HFD and sex differences on glycerolipids and CE in the renal cortex

To better investigate the effects of sex and diet on the kidney lipidome, lipidomic data for males and females were presented based on the number of double bonds and acyl chain length for each lipid class. Neutral lipid accumulation in kidneys is associated with diabetic nephropathy and obesity-induced CKD [20]. We evaluated DG, TG, and CE accumulation, as well as their FA-acid chain composition, in males and females fed a HFD. As illustrated in Fig. 4, both sexes displayed similar accumulation of glycerolipids (DG and TG). Analysis of acyl chain length and unsaturated bonds demonstrated the greatest accumulation of TG and DG with unsaturated 18:1 and 18:2 in HFD-fed mice. Moreover, DG 22:6 was also increased in HFD-fed male and female mice (Fig. 4A-D). Only females exhibited a significant increase in 16:0 for DG and TG lipid classes (Fig. 4B). Regarding sterol lipid accumulation, the data showed that renal CE content was lower in females than in males, even in lean animals (Fig. 4E, F). HFD induces increased levels of 18:2 and 20:4 CE, with a more marked accumulation in males. Collectively, these data demonstrate that males and females do



Fig. 4 Sex differences in lipid content profiles (DG, TG, and CE) in renal cortex tissue from males and females in response to a HFD. Sum of the quantitative amount of FA species within a class after normalization (DNA content from tissue samples). **A, B** Changes in DG profile in males and females fed a HFD or a LFD. **C, D** Changes in TG profile in males and females fed a HFD or a LFD. **E, F** Changes in CE profile in males and females fed a HFD or a LFD. Statistical analyses were performed using multiple *t*-tests with *p* values corrected for multiple comparison using the Bonferroni method. Data are presented as means ± SEM. *n* = 4–6 in each group

not exhibit a drastic differential response in neutral lipid accumulation when fed a HFD.

Sex-specific phospholipid accumulation in the renal cortex in response to a HFD and associated autophagy dysfunction

Using an in vitro model of renal lipid overload, Yamamoto et al. demonstrated impaired lysosomal accumulation and excessive accumulation of phospholipids originating from the plasma and organelle membranes [22]. Lipid-omics analysis also revealed differences in phospholipid accumulation between the sexes and diets. As illustrated in Fig. 5A, B, females were remarkably protected from PC accumulation in the renal tissues. While LFD-fed males and females presented the same FA profile for PC, HFD males presented a significant increase in PC content, mostly for palmitic acid (16:0) but also 14:1, and fatty acids with 18C. In contrast, the FA profiles of other phospholipid classes, such as PE, PG, and PS, did not show any difference between males and females (Fig. 5C–H).

А

AND

bm/

_مد (

Indeed, Fig. 5C, D shows the accumulation of PE in males and females in response to HFD. Both HFD-fed males and females accumulated PE 16:0, PE with 18C, 20:4 and 22:6 with similar fold changes between sexes.

Studies suggest that HFD induces lysosomal storage and autophagy dysfunction in male mice, similar to the intra-lysosomal accumulation of phospholipids observed in PTC of animals and humans exposed to aminoglycosides or lysosomal storage diseases [13, 22, 41]. To assess this phenotype, we evaluated the abundance of the lysosomal marker LAMP-1 in renal tissues from male and female mice subjected to a LFD or HFD. A more pronounced LAMP-1 staining in renal cortex was observed in HFD-fed males, particularly in the PTC at the site of lipid droplet accumulation (Fig. 6A). The percentage of LAMP-1-positive area was significantly increased in HFD-fed males compared to LFD-fed males, but not in females with a significant sex \times diet interaction (P_{Int} < 0.05), suggesting that lysosome accumulation in the renal cortex is sex-specific



В

DNA

pm/lom

LFD female



Fig. 6 Sex differences in renal lysosomal dysfunction in males and females in response to a HFD. **A** Representative photomicrograph (original magnification \times 40) showing LAMP1–positive staining on intracellular vacuoles in males and females fed a HFD or a LFD. **B** Related quantitative analysis of LAMP1-positive staining. **C–E** Real-time quantitative PCR analysis for LAMP1, Cathepsin D and p62 genes. mRNA expression was performed with kidney tissue from LFD and HFD female and male mice and normalized against 18S. Statistical analyses were performed using two-way ANOVA followed by Tukey post-test. *p < 0.05; **p < 0.01. Data are presented as means ± SEM

(Fig. 6B). Moreover, the expression of *Lamp1*, *Cathepsin D*, and *p62* was higher in males than in females, with a significant sex×diet interaction ($P_{Int} < 0.05$) for *Lamp1*, as shown in Fig. 6C–E. These data suggest the dysregulation of lysosomal markers (as described in [22]) in a sex-specific manner.

Sex-specific CER accumulation in the renal cortex in response to HFD is associated with adiponectin and AMPK signaling

Sphingolipids such as CER are mainly produced by the endoplasmic reticulum via de novo synthesis [42]. Among the sphingolipid classes, SM accumulation has been implicated in metabolic disturbances in diabetic nephropathy [43]. Male and female mice present differential renal accumulation of SM and CER species in response to HFD. While a strong accumulation of 16:0, 24:0, and 24:1 SM was observed in males, females only displayed moderate accumulation of 16:0 (Fig. 7A, B). CER accumulation in the kidneys is well known to contribute to renal disorders. Here, long chain (C16) and very-long-chain (22 to 24C) CER were particularly increased in HFD-fed male mice, while females exhibited a significant accumulation of 16:0 but no difference in very-long-chain CER content (Fig. 7C, D). The mRNA expression levels of CER synthases CerS5 and Cers6, which predominantly produce C16-CER, were higher in LFD males than in LFD females. Their expression was significantly upregulated in both sexes fed a HFD (Fig. 7E, F). In contrast, the relative expression of CerS2, which produces CER incorporating 22- to 24-carbon FA, was only significantly increased in male HFD-fed mice. CER can be deacylated by a family of ceramidases that produce sphingosines to prevent lipotoxicity. Therefore, the expression of alkaline ceramidases *Acer2* and *Acer3* was quantified and was found to be upregulated in males but not in females in response to HFD.

An interesting link has recently been found between adiponectin levels and CER accumulation in HFD-fed mice, suggesting that adiponectin may protect against the accumulation of CER [44]. Therefore, based on our previous studies demonstrating that renal lipotoxicity in HFD-fed mice was associated with decreased AMPK activity, we further investigated the adiponectin-AMPK axis. Here, P-AMPK to AMPK ratios were first investigated in the kidneys of LFD and HFD male and female mice. The evaluation of AMPK activity in renal tissue by Western blotting was performed by calculating the ratio between phosphorylated (thr172) AMPKa and total AMPKa. As expected, a decreased P-AMPK:AMPK ratio was observed in HFD-fed male mice compared with LFD, reflecting a decrease in AMPK activity (Fig. 8A, B). Interestingly, female mice showed higher AMPK activity than males ($P_{Sex} < 0.0001$) without a decrease in P-AMPK:AMPK ratio when fed with HFD. Adiponectin stimulates the activation of AMPK via AdipoR1, and adiponectin-deficient mice show decreased AMPK phosphorylation [45]. In this regard, we evaluated the adiponectin/AMPK pathway in males and females. Plasma adiponectin levels were



Fig. 7 Sex differences in sphingolipid content profile and CER metabolism in renal cortex tissue from males and females in response to a HFD. Sum of the quantitative amount of FA species within a class after normalization (DNA content from tissue samples). **A, B** Changes in SM profile in males and females fed a HFD or a LFD. **C, D** Changes in CER profile in males and females fed a HFD or a LFD. Statistical analyses were performed using multiple t-tests with P values corrected for multiple comparison using the Bonferroni method. Data are presented as means \pm SEM. n=4-6 in each group. **E–F** Real-time quantitative PCR analysis for *CerS2, CerS6, Acer2* and *Acer3* genes. mRNA expression was performed with kidney tissue from LFD and HFD female and male mice and normalized against 18S. Statistical analyses were performed using two-way ANOVA followed by Tukey post-test. *p < 0.05; **p < 0.01; ***p < 0.001. Data are presented as means \pm SEM

not significantly affected by HFD, but female mice presented higher levels of adiponectin than males ($P_{Sex} < 0.005$). This was significantly correlated with AMPK activity (phospho-AMPK to AMPK ratio) in renal tissues (Fig. 8D). Moreover, HFD selectively decreased AdipoR1 mRNA and protein expression in males (Fig. 8E–G). Taken together, these data suggest dysregulation of the adiponectin/AMPK pathway in HFD-fed males but not in females.

Discussion

The U.S. National Institutes of Health (NIH) initiated recommendations to address over-reliance on males in preclinical research and improve sex balance [46]. Underrepresentation of female subjects in experimental and clinical research still results in a lack of understanding of the basic biological processes related to sex and poorer treatment outcomes for women [47]. Numerous studies have demonstrated that males and females differ in their physiology and metabolic homeostasis [48–50]. HFD-fed mice have been widely used as a suitable model to study obesity-related disorders [51]. While obesity-induced kidney disease is now well characterized in HFD-fed C57Bl/6J male mice, there are still no reliable data on the renal consequences of obesity in females. In this study, we explored the sex-specific effects of a HFD on kidney dysfunction and renal lipid accumulation. Indeed, we have previously demonstrated that dysregulated renal lipid metabolism and associated lipotoxicity play a major role in CKD development and progression in HFD-fed male mice [12–15]. Moreover, previous studies have demonstrated sex-specific lipid molecular signatures in the adipose tissue and liver of obese mice, contributing to the sex-dependent response to obesity [52]. Through targeted lipidomic analysis, we further characterized the renal lipid species composition, which may contribute to sexual dimorphism in renal disease development.

Despite similar BW gain and increased circulating leptin concentrations in both males and females, HFD feeding induced sex-related differences in metabolic parameters. Males but not females develop insulin resistance with hyperglycemia and hyperinsulinemia. This is consistent with other studies demonstrating that high caloric intake does not affect insulin sensitivity in female [53–55]. Importantly, female mice exposed to a prolonged period of HFD (36 weeks) have been associated with delayed metabolic dysfunction and insulin resistance compared to males [4]. Protection against



Fig. 8 Sex differences in the renal AMPK/adiponectin pathway in males and females in response to a HFD. **A** Representative Western blotting of total and P-AMPK levels in renal cortex tissue. β-Actin was used as internal control. **B** Relative densitometry of the immunoblots representing P-AMPK to AMPK ratio in each group. β-actin was used to illustrate protein loading. **C** Plasma adiponectin level in males and females fed a HFD or a LFD at week 16. **D** Correlation between plasma adiponectin and AMPK activity (P-AMPK to AMPK ratio) for all animals. A significant positive linear correlation is present. **E** Representative Western blotting of AdipoR1 level in renal cortex tissue. β-actin was used as internal control. **F** Relative densitometry of the immunoblots representing AdipoR1 in each group. **G** Real-time quantitative PCR analysis for AdipoR1 gene. mRNA expression was performed with renal cortex tissue from LFD and HFD female and male mice and normalized against 18S. Statistical analyses were performed using two-way ANOVA followed by Tukey post-test. ***p* < 0.01. Data are presented as means ± SEM

obesity-associated insulin resistance was also decreased in ovariectomized females, suggesting that sex hormones play a crucial role in protecting female mice from metabolic disturbances induced by obesity [56]. Clinical studies have also demonstrated that diabetes is more prevalent in obese men than in obese women [57]. However, menopause has been associated with detrimental effects on glucose homeostasis, while estrogen-based hormonal therapy protects against the development of diabetes [58]. There is clear evidence in clinical and experimental studies that the development of insulin resistance in obese individuals is influenced by sex and affects more men than women [57]. However, the relationship between sex and obesity-related disorders, such as dyslipidemia, ectopic lipid accumulation, and lipotoxicity, remains ambiguous. Dyslipidemia is a common feature of obesity, and is associated with diabetes and fatty liver disease [59]. In the present study, both male and female HFD-fed mice presented increased plasma lipid levels, with a more pronounced increase in plasma FFA in females, whereas males showed higher levels of plasma cholesterol. Studies have shown that obesity increases plasma FFA levels in both sexes due to increased lipolysis in adipose tissue [60]. However, female animals as well as women are protected from FFA-induced insulin resistance [61, 62]. Experimental data from animal models suggest that estrogens protect insulin-sensitive tissues from insulin resistance by activating the ER α pathway in females [63]. Many studies have reported sex differences in diet-induced non-alcoholic fatty liver disease development, with varying outcomes depending on the strain or diet composition [64-67]. Here, increased plasma lipid levels were associated with hepatic steatosis in both the sexes. However, hepatic steatosis was more pronounced in males fed a HFD than in females, which is relevant compared to experimental and epidemiological studies [67, 68].

Epidemiological studies have found that sex and gender differences exist in the susceptibility, prevalence, and progression of CKD [10]. Although premenopausal women appear to be protected from non-diabetic renal diseases, the incidence of CKD is higher in women than in men, whereas progression to end-stage renal disease is more common in men [69]. Many experimental models of kidney diseases have demonstrated that sex hormones in females exert renoprotective effects [70–72]. Regarding obesity, the relationship between body mass index and CKD also seems to differ according to gender [73]. The lipid metabolism of males and females significantly differs, which may impact adaptation to lipid overload in targeted organs during obesity, including the kidneys [50].

In the present study, we showed that female mice fed a HFD were protected from renal function impairments. Indeed, hallmarks of renal dysfunction, such as increased albuminuria, proteinuria, and even increased levels of urine hydrogen peroxide, were significantly elevated in obese male mice, but not in females. We and others have previously demonstrated that the proximal tubule is the primary site of lipid deposition during obesity [13, 14, 22]. Concentric membrane layer structures (multilamellar bodies) as well as cytoplasmic neutral lipid droplets have been found in murine models of obesity-induced CKD, as well as in patients [13, 20, 22, 74]. In particular, our previous study demonstrated that the proximal tubular epithelium of male mice fed a HFD accumulated large lipid vacuoles [12]. Interestingly, in the current study, no vacuolization was detected in the renal cortex of the female mice. Therefore, to better link renal physiopathology to lipid species composition between males and females, a targeted lipidomic approach was used. Although both male and female mice fed a HFD exhibited renal lipid accumulation, male mice were found to be more affected than females for all lipid species, with a drastic change in renal lipid composition.

Importantly, lipidomic analysis showed that females do accumulate specific lipid species when fed with HFD, despite the absence of vascularized tubules in renal cortex sections. This is particularly true for neutral lipids, such as DG, TG, and cholesterol. Indeed, a very similar accumulation of DG and TG, with mostly side chains of oleic (18:1) and linoleic (18:2) acids, was found in both males and females fed a HFD. In addition, kidneys of obese mice mostly accumulate CE with linoleic and arachidonic acid, while males present higher levels of CE compared to females. This suggests that HFD feeding leads to renal lipid accumulation in both sexes but not necessarily lipotoxicity. Indeed, lipotoxicity is a consequence of an imbalance between FA uptake, utilization (notably through beta-oxidation), and storage. Storage relies on esterification of a glycerol backbone to form TG, which is then stored within lipid droplets. Triglycerides themselves are not toxic but serve as an active reservoir of FA as a direct consequence of lipid overload. Renal cells are known to have an important capacity to store TG as lipid droplets, notably during fasting, in both males and females [75]. Increased serum NEFA levels are directly associated with increased TG accumulation in the kidneys. Thus, increased lipid storage in the kidneys of male and female mice could be the consequence of elevated plasma levels of NEFA found in both sexes when fed with HFD. Interestingly, increased levels of DG were not associated with kidney impairment in females, suggesting that DG intermediate accumulation in the kidney could not be particularly linked to direct lipotoxicity.

Besides, caloric excess leads to upregulation of the de novo CER synthesis pathway [76]. CER play an important role as lipotoxic mediators of metabolic dysfunction in fatty liver disease, cardiovascular diseases, and diabetes [77]. Differences in circulating CER profiles have been significantly associated with the development of macroalbuminuria and CKD in humans [78]. However, the relevance of CER metabolism in kidneys remains unclear. Here, we demonstrated sex differences in the abundance and composition of CER lipid species. While HFD-fed male mice displayed a strong accumulation of long chain (C16) as well as very-long-chain (C22-C24)-CER, females only showed a moderate but still significant increase in C16-CER levels. Numerous studies have suggested that specific CER species are more relevant in pathological events than changes in total CER concentrations, although the exact role of each CER species is still debated [76]. Law et al. demonstrated that very long-chain CER, but not long-chain (C16 and C18) CER, causes mitochondrial dysfunction and cell death in cardiomyocytes [79]. In our study, only male HFDfed mice presented an increased level of very-long-chain CER and exhibited evidence of renal dysfunction (as attested by marked albuminuria, proteinuria, and urine hydrogen peroxide). Moreover, the expression of CerS2, which produces very long-chain (C20-26) CER species, was only enhanced in male mice fed a HFD. In contrast, the expression of CerS5 and Cers6, which promote the incorporation of long-chain (e.g., C16) acyl groups, was increased in both sexes and is related to increased levels of C16-CER. Collectively, these data suggest that HFD-fed females are protected from very-long-chain CER accumulation in the kidneys and their associated lipotoxic effects. Recently, a link between adiponectin axis and CER metabolism was highlighted, suggesting that adiponectin, through its interaction with its receptors, induces the activation of ceramidases that lead to beneficial effects on mitochondria in adipocytes and hepatocytes [26]. Adiponectin is a 30-kDa circulating plasma protein that is primarily secreted by adipocytes. Adiponectin can bind to three receptors: AdipoR1, -2 and T-cadherin. Adipo-R1 and -R2 have anti-atherogenic, anti-inflammatory, and hypoglycemic properties [80]. Here, even though genic ceramidase expression was not increased in females, plasma adiponectin levels as well as mRNA and protein levels of the adiponectin receptor were higher in HFD-fed females than in males. Adiponectin is a key regulator of glucose and lipid metabolism in the kidneys [81]. This adipokine is involved in the reduction of inflammation, fibrosis, and oxidative stress through activation of AMPK and peroxisome proliferator-activated receptor alpha (PPAR α) [82]. Besides, it has been demonstrated that AMPK inhibits de novo CER synthesis in skeletal muscle and astrocytes [83, 84]. AMPKα1 deficiency was also associated with lipid droplet accumulation in the kidney tubules and increased CER accumulation [85]. Moreover, we previously demonstrated the crucial role of AMPK in renal cell dysfunction [13, 14]. Female mice were protected from decreased AMPK activity, which correlated with higher circulating levels of adiponectin in females. Studies using AdipoR1 knockout mice showed decreased AMPK activity, while AMPK activity correlated with adiponectin levels in an obesity model [45, 86]. Moreover, treatment of db/db mice with an activator of adiponectin, AdipoRon, resulted in upregulation of phosphorylated AMPK in the kidney along with reduced inflammation and lipotoxicity [26]. Thus, a dysregulated adiponectin/AMPK pathway in HFD-fed males could lead to CER-induced lipotoxicity and renal injury. AMPK plays a key role in the regulation of autophagy in mammalian cells [87]. Alterations in autophagy, lysosomal dysfunction, and lysosomal lipid storage have been described as the hallmarks of renal lipotoxicity [3]. Yamamoto et al. demonstrated the accumulation of dysfunctional lysosomes containing phospholipids of plasma and organelle membrane origin using an in vitro model of renal lipid overload [22]. In the present study, lipidomic analysis revealed significant accumulation of PC species in the kidneys of HFD-fed male mice. This change was associated with an increase in the number of lipid vacuoles in proximal tubules that were positive for lysosomal markers. Therefore, these results are in line with those of Yamamoto et al. and confirm that the intra-lysosomal lipid storage found in males is likely composed of phospholipids [22]. Moreover, consistent with our previous studies, our results showed that maintaining AMPK activity, as observed in female mice, is associated with preserved tubular homeostasis.

Perspectives and significance

Our study revealed that renal lipid species signatures in mice in response to HFD are markedly different between sexes. Our results showed that females are protected from the accumulation of very long CER, which might contribute to renal lipotoxicity in males. To further advance our understanding of the specific role of CER metabolism in response to HFD, future studies targeting ceramide synthase and ceramidase in the mouse kidney would be highly valuable. Examining ceramidase activity in the kidney could also provide additional insights into the mechanisms underlying the protection of female mice from very-long-chain ceramide accumulation. In addition, only males presented a strong accumulation of phospholipids related to autophagy and lysosomal impairment. Furthermore, we explored the putative link between the adiponectin-AMPK axis and the prevention of lipotoxicity and CKD development in females. Further studies are warranted to delineate the precise role of adiponectin in female protection against renal lipotoxicity. Indeed, it cannot be excluded that the sustained AMPK activity in obese females may be attributed to factors other than adiponectin and that the low levels of toxic lipids such as ceramides may play a beneficial role. Thus, the use of adiponectin knockout female mice on HFD

would be an interesting mechanistic strategy to confirm the adiponectin/AMPK axis in renal tissue. Moreover, studies using human samples such as renal biopsy of diabetic/obese patients as well as related clinical data are needed to enhance the clinical relevance of our findings and provide valuable insights in the sexual dimorphism of obesity-induced renal lipid accumulations.

This novel characterization might contribute to the understanding of functional differences in obesityinduced CKD between males and females.

Acknowledgements

The authors thank Jean-François Gaussin for technical support.

Author contributions

Among the authors, FJ, NC and A-ED conceived the experiments; FJ, LP, MD, IJ and BM performed the experiments and interpreted the data; JD generated lipidomics data. OB provided the technical assistance; A-ED, NC and JS oversaw the experiments which were performed in their respective laboratories; FJ and A-ED prepared the manuscript. All authors approved the final version of the manuscript and agree to be accountable for all aspects of the work. All persons designated as authors qualify for authorship, and all those who qualify for authorship are listed. All authors have read and agreed to the published version of the manuscript.

Funding

This work was supported by Grants from the UMONS Research Institute for Health Sciences and Technology (Belgium), the FRMH (Fonds pour la Recherche Médicale dans le Hainaut, Belgium), the French Diabetes Society and the Fonds de la Recherche Scientifique (FNRS).

Availability of data and materials

Please contact author for data requests.

Declarations

Ethics approval and consent to participate Not applicable.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Laboratory of Metabolic and Molecular Biochemistry, Faculty of Medicine and Pharmacy, Research Institute for Health Sciences and Technology, University of Mons (UMONS), Mons, Belgium. ²Molecular Physiology Research Unit (URPhyM), Namur Research Institute for Life Sciences (NARILIS), University of Namur (UNamur), Namur, Belgium. ³Biochemistry and Cellular Biology Research Unit (URBC), Namur Research Institute for Life Sciences (NARILIS), University of Namur (UNamur), Namur, Belgium. ⁴Laboratory of Lipid Metabolism and Cancer, Department of Oncology, KU Leuven, Leuven, Belgium.

Received: 16 April 2023 Accepted: 4 September 2023 Published online: 28 September 2023

References

- Blüher M. Obesity: global epidemiology and pathogenesis. Nat Rev Endocrinol. 2019;15:288–98.
- Garofalo C, et al. A systematic review and meta-analysis suggests obesity predicts onset of chronic kidney disease in the general population. Kidney Int. 2017;91:1224–35.

- Juszczak F, Caron N, Mathew AV, Declèves A-E. Critical role for AMPK in metabolic disease-induced chronic kidney disease. Int J Mol Sci. 2020;21:7994.
- Elzinga SE et al. Sex differences in insulin resistance, but not peripheral neuropathy, in a diet-induced prediabetes mouse model. Dis Model Mech 2021;14.
- Chen K-HE, Lainez NM, Coss D. Sex differences in macrophage responses to obesity-mediated changes determine migratory and inflammatory traits. J Immunol. 2021;206:141–53.
- Maric I, et al. Sex and species differences in the development of dietinduced obesity and metabolic disturbances in rodents. Front Nutr. 2022;9: 828522.
- Chukijrungroat N, Khamphaya T, Weerachayaphorn J, Songserm T, Saengsirisuwan V. Hepatic FGF21 mediates sex differences in high-fat high-fructose diet-induced fatty liver. Am J Physiol Endocrinol Metab. 2017;313:E203–12.
- 8. Wang X-H, et al. Women are at a higher risk of chronic metabolic diseases compared to men with increasing body mass index in China. Front Endocrinol (Lausanne). 2020;11:127.
- Cobo G, et al. Sex and gender differences in chronic kidney disease: progression to end-stage renal disease and haemodialysis. Clin Sci. 2016;130:1147–63.
- Carrero JJ, Hecking M, Chesnaye NC, Jager KJ. Sex and gender disparities in the epidemiology and outcomes of chronic kidney disease. Nat Rev Nephrol. 2018;14:151–64.
- 11. Wong SK, Chin K-Y, Suhaimi FH, Fairus A, Ima-Nirwana S. Animal models of metabolic syndrome: a review. Nutr Metab Lond. 2016;13:65.
- Declèves A-E, Mathew AV, Cunard R, Sharma K. AMPK mediates the initiation of kidney disease induced by a high-fat diet. J Am Soc Nephrol. 2011;22:1846–55.
- Declèves A-E, et al. Regulation of lipid accumulation by AMKactivated kinase in high fat diet–induced kidney injury. Kidney Int. 2014;85:611–23.
- 14. Juszczak F, et al. Delayed exercise training improves obesity-induced chronic kidney disease by activating AMPK pathway in high-fat diet-fed mice. Int J Mol Sci. 2020;22:350.
- Declèves A-E, et al. AMP-activated protein kinase activation ameliorates eicosanoid dysregulation in high-fat-induced kidney disease in mice. J Lipid Res. 2019;60:937–52.
- D'Agati VD, et al. Obesity-related glomerulopathy: clinical and pathologic characteristics and pathogenesis. Nat Rev Nephrol. 2016;12:453–71.
- 17. Deji N. et al. Structural and functional changes in the kidneys of high-fat diet-induced obese mice.
- Declèves A-E, et al. Lipid signaling and lipotoxicity in metaflammation: indications for metabolic disease pathogenesis and treatment. J Lipid Res. 2019;60:2099–114.
- Jiang T, et al. Diet-induced obesity in C57BL / 6J mice causes increased renal lipid accumulation and glomerulosclerosis via a sterol regulatory element-binding protein-1c-dependent pathway. J Biol Chem. 2005;280:32317–25.
- Herman-Edelstein M, Scherzer P, Tobar A, Levi M, Gafter U. Altered renal lipid metabolism and renal lipid accumulation in human diabetic nephropathy. J Lipid Res. 2014;55:561–72.
- Kang HM, et al. Defective fatty acid oxidation in renal tubular epithelial cells plays a key role in kidney fibrosis development HHS Public Access. Nat Med. 2015;21:37–46.
- Yamamoto T, Takabatake Y, Takahashi A, Kimura T. High-fat diet-induced lysosomal dysfunction and impaired autophagic flux contribute to lipotoxicity in the kidney. J Am Soc Nephrol. 2016.
- Rampanelli E, et al. Excessive dietary lipid intake provokes an acquired form of lysosomal lipid storage disease in the kidney. J Pathol. 2018;246:470–84.
- Li L, et al. Metabolomics reveal mitochondrial and fatty acid metabolism disorders that contribute to the development of DKD in T2DM patients. Mol Biosyst. 2017;13:2392–400.
- Dugan LL, et al. AMPK dysregulation promotes diabetes-related reduction of superoxide and mitochondrial function. J Clin Invest. 2013;123:4888–99.

- Choi SR, et al. Adiponectin receptor agonist AdipoRon decreased ceramide, and lipotoxicity, and ameliorated diabetic nephropathy. Metabolism. 2018;85:348–60.
- 27. Lee M-J, et al. A role for AMP-activated protein kinase in diabetes-induced renal hypertrophy. Am J Physiol Renal Physiol. 2007;292:F617–27.
- Eid AA, et al. AMP-activated protein kinase (AMPK) negatively regulates Nox4-dependent activation of p53 and epithelial cell apoptosis in diabetes. J Biol Chem. 2010;285:37503–12.
- Han SH, et al. Deletion of Lkb1 in renal tubular epithelial cells leads to CKD by altering metabolism. J Am Soc Nephrol. 2016;27:439–53.
- Herzig S, Shaw RJ. AMPK: guardian of metabolism and mitochondrial homeostasis. Nat Rev Mol Cell Biol. 2017. https://doi.org/10.1038/nrm. 2017.95.
- Hasanvand A, et al. Metformin attenuates streptozotocin-induced diabetic nephropathy in rats through activation of AMPK signaling pathway. J Nephropathol. 2018;7:37–42.
- Zhou X, et al. PAN-AMPK activation improves renal function in a rat model of progressive diabetic nephropathy. J Pharmacol Exp Ther. 2019;371:45–55.
- Kubota N, et al. Adiponectin stimulates AMP-activated protein kinase in the hypothalamus and increases food intake. Cell Metab. 2007;6:55–68.
- Sharma K, et al. Adiponectin regulates albuminuria and podocyte function in mice. J Clin Invest. 2008;118:1645–56.
- Kim Y, et al. The adiponectin receptor agonist adiporon ameliorates diabetic nephropathy in a model of type 2 diabetes. J Am Soc Nephrol. 2018;29:1108–27.
- Yamauchi T, et al. Adiponectin stimulates glucose utilization and fattyacid oxidation by activating AMP-activated protein kinase. Nat Med. 2002;8:1288–95.
- Saltevo J, Kautiainen H, Vanhala M. Gender differences in adiponectin and low-grade inflammation among individuals with normal glucose tolerance, prediabetes, and type 2 diabetes. Gend Med. 2009;6:463–70.
- Song HJ, et al. Gender differences in adiponectin levels and body composition in older adults: Hallym aging study. BMC Geriatr. 2014;14:8.
- Ryu J-E, et al. Evaluation of nonalcoholic fatty liver disease in C57BL/6J mice by using MRI and histopathologic analyses. Comp Med. 2015;65:409–15.
- Perakakis N, Farr OM, Mantzoros CS. Leptin in leanness and obesity: JACC state-of-the-art review. J Am Coll Cardiol. 2021;77:745–60.
- Platt FM, Boland B, van der Spoel AC. Lysosomal storage disorders: the cellular impact of lysosomal dysfunction. J Cell Biol. 2012;199:723–34.
- Bhat OM, Yuan X, Li G, Lee R, Li P-L. Sphingolipids and redox signaling in renal regulation and chronic kidney diseases. Antioxid Redox Signal. 2018;28:1008–26.
- Zhang G, Darshi M, Sharma K. The Warburg effect in diabetic kidney disease. Semin Nephrol. 2018;38:111–20.
- Holland WL, et al. Inducible overexpression of adiponectin receptors highlight the roles of adiponectin-induced ceramidase signaling in lipid and glucose homeostasis. Mol Metab. 2017;6:267–75.
- 45. Niu M, et al. Adiponectin induced AMP-activated protein kinase impairment mediates insulin resistance in Bama mini-pig fed high-fat and high-sucrose diet. Asian-Australasian J Animal Sci. 2017;30:1190–7.
- Sandberg K, Umans JG. Recommendations concerning the new U.S. National Institutes of Health initiative to balance the sex of cells and animals in preclinical research. FASEB J Biol. 2015;29:1646–52.
- 47. Beery AK. Inclusion of females does not increase variability in rodent research studies. Curr Opin Behav Sci. 2018;23:143–9.
- Mauvais-Jarvis F. Sex differences in metabolic homeostasis, diabetes, and obesity. Biol Sex Differ. 2015;6:14.
- Kautzky-Willer A, Harreiter J, Pacini G. Sex and gender differences in risk, pathophysiology and complications of type 2 diabetes mellitus. Endocr Rev. 2016;37:278–316.
- 50. Palmer BF, Clegg DJ. The sexual dimorphism of obesity. Mol Cell Endocrinol. 2015;402:113–9.
- de Moura e Dias M, et al. Diet-induced obesity in animal models: points to consider and influence on metabolic markers. Diabetol Metab Syndr. 2021;13:32.
- González-Granillo M, et al. Sex-specific lipid molecular signatures in obesity-associated metabolic dysfunctions revealed by lipidomic characterization in ob/ob mouse. Biol Sex Differ. 2019;10:11.

- Pettersson US, Waldén TB, Carlsson P-O, Jansson L, Phillipson M. Female mice are protected against high-fat diet induced metabolic syndrome and increase the regulatory T cell population in adipose tissue. PLoS ONE. 2012;7: e46057.
- Glavas MM, et al. Developmental timing of high-fat diet exposure impacts glucose homeostasis in mice in a sex-specific manner. Diabetes. 2021;70:2771–84.
- Casimiro I, Stull ND, Tersey SA, Mirmira RG. Phenotypic sexual dimorphism in response to dietary fat manipulation in C57BL/6J mice. J Diabetes Complications. 2021;35: 107795.
- Bader J, et al. High-fat diet-fed ovariectomized mice are susceptible to accelerated subcutaneous tumor growth potentially through adipose tissue inflammation, local insulin-like growth factor release, and tumor associated macrophages. Oncotarget. 2020;11:4554–69.
- Nordström A, Hadrévi J, Olsson T, Franks PW, Nordström P. Higher prevalence of type 2 diabetes in men than in women is associated with differences in visceral fat mass. J Clin Endocrinol Metab. 2016;101:3740–6.
- Kanaya AM, et al. Glycemic effects of postmenopausal hormone therapy: the Heart and Estrogen/progestin Replacement Study. A randomized, double-blind, placebo-controlled trial. Ann Intern Med. 2003;138:1–9.
- Jung UJ, Choi M-S. Obesity and its metabolic complications: the role of adipokines and the relationship between obesity, inflammation, insulin resistance, dyslipidemia and nonalcoholic fatty liver disease. Int J Mol Sci. 2014;15:6184–223.
- Samuel P, Khan MA, Nag S, Inagami T, Hussain T. Angiotensin AT(2) receptor contributes towards gender bias in weight gain. PLoS ONE. 2013;8: e48425.
- 61. Palmisano BT, Zhu L, Eckel RH, Stafford JM. Sex differences in lipid and lipoprotein metabolism. Mol Metab. 2018;15:45–55.
- 62. Frias JP, et al. Decreased susceptibility to fatty acid-induced peripheral tissue insulin resistance in women. Diabetes. 2001;50:1344–50.
- 63. Hevener A, Reichart D, Janez A, Olefsky J. Female rats do not exhibit free fatty acid-induced insulin resistance. Diabetes. 2002;51:1907–12.
- 64. Spruss A, et al. Female mice are more susceptible to nonalcoholic fatty liver disease: sex-specific regulation of the hepatic AMP-activated protein kinase-plasminogen activator inhibitor 1 cascade, but not the hepatic endotoxin response. Mol Med. 2012;18:1346–55.
- Gasparin FRS, et al. Sex differences in the development of hepatic steatosis in cafeteria diet-induced obesity in young mice. Biochim Biophys Acta Mol Basis Dis. 2018;1864:2495–509.
- Lee Y-H, et al. Sex-specific metabolic interactions between liver and adipose tissue in MCD diet-induced non-alcoholic fatty liver disease. Oncotarget. 2016;7:46959–71.
- 67. de Conti A, et al. Characterization of the variability in the extent of nonalcoholic fatty liver induced by a high-fat diet in the genetically diverse Collaborative Cross mouse model. FASEB J. 2020;34:7773–85.
- Lonardo A, et al. Sex differences in nonalcoholic fatty liver disease: state of the art and identification of research gaps. Hepatology. 2019;70:1457–69.
- 69. Bairey Merz CN, et al. Sex and the kidneys: current understanding and research opportunities. Nat Rev Nephrol. 2019;15:776–83.
- Irsik DL, et al. Renoprotective impact of estrogen receptor-α and its splice variants in female mice with type 1 diabetes. Am J Physiol Renal Physiol. 2018;315:F512–20.
- Singh AP, Singh N, Pathak D, Bedi PMS. Estradiol attenuates ischemia reperfusion-induced acute kidney injury through PPAR-γ stimulated eNOS activation in rats. Mol Cell Biochem. 2019;453:1–9.
- 72. Aufhauser DDJ, et al. Improved renal ischemia tolerance in females influences kidney transplantation outcomes. J Clin Invest. 2016;126:1968–77.
- Komura H, Nomura I, Kitamura K, Kuwasako K, Kato J. Gender difference in relationship between body mass index and development of chronic kidney disease. BMC Res Notes. 2013;6:463.
- 74. Sharma SG, et al. The modern spectrum of renal biopsy findings in patients with diabetes. Clin J Am Soc Nephrol. 2013;8:1718–24.
- 75. Scerbo D, et al. Kidney triglyceride accumulation in the fasted mouse is dependent upon serum free fatty acids. J Lipid Res. 2017;58:1132–42.
- Nicholson RJ, Pezzolesi MG, Summers SA. Rotten to the cortex: ceramidemediated lipotoxicity in diabetic kidney disease. Front Endocrinol Lausanne. 2020;11:622692.
- 77. Chaurasia B, Summers SA. Ceramides in metabolism: key lipotoxic players. Annu Rev Physiol. 2021;83:303–30.

- Klein RL, et al. Decreased plasma levels of select very long chain ceramide species are associated with the development of nephropathy in type 1 diabetes. Metabolism. 2014;63:1287–95.
- Law BA, et al. Lipotoxic very-long-chain ceramides cause mitochondrial dysfunction, oxidative stress, and cell death in cardiomyocytes. FASEB J. 2018;32:1403–16.
- Fang H, Judd RL. Adiponectin regulation and function. Compr Physiol. 2018;8:1031–63.
- Przybyciński J, Dziedziejko V, Puchałowicz K, Domański L, Pawlik A. Adiponectin in chronic kidney disease. Int J Mol Sci. 2020;21:9375.
- 82. Kim J, Yang G, Kim Y, Kim J, Ha J. AMPK activators: mechanisms of action and physiological activities. Exp Mol Med. 2016;48:e224–312.
- Blázquez C, Geelen MJ, Velasco G, Guzmán M. The AMP-activated protein kinase prevents ceramide synthesis de novo and apoptosis in astrocytes. FEBS Lett. 2001;489:149–53.
- Erickson KA, et al. AICAR inhibits ceramide biosynthesis in skeletal muscle. Diabetol Metab Syndr. 2012;4:45.
- Ma H, et al. Dephosphorylation of AMP-activated protein kinase exacerbates ischemia/reperfusion-induced acute kidney injury via mitochondrial dysfunction. Kidney Int. 2022;101:315–30.
- Yamauchi T, et al. Targeted disruption of AdipoR1 and AdipoR2 causes abrogation of adiponectin binding and metabolic actions. Nat Med. 2007;13:332–9.
- Alers S, Löffler AS, Wesselborg S, Stork B. Role of AMPK-mTOR-Ulk1/2 in the regulation of autophagy: cross talk, shortcuts, and feedbacks. Mol Cell Biol. 2012;32:2–11.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

